Author's response to reviews

Title: Esomeprazole for the treatment of erosive esophagitis in children: an international, multicenter, randomized, parallel-group, double-blind (for dose) study

Authors:

Vasuhandra Tolia (vasu.tolia@gmail.com)
Nader N Youssef (Nader.Youssef@ahsys.org)
Mark A Gilger (mgilger@bcm.edu)
Barry Traxler (barry.traxler@astrazeneca.com)
Marta Illueca (marta.illueca@astrazeneca.com)

Version: 3 Date: 11 May 2010

Author's response to reviews: see over
May 13, 2010

BMC Series Editorial Production Team
Tel: +44 (0) 20 3192 2014
Email: bmcseriesprepublication@biomedcentral.com

Dear BMC Series Editorial Production Team:

Thank you for your review of our manuscript titled “Esomeprazole for the treatment of erosive esophagitis in children: an international, multicenter, randomized, parallel-group, double-blind (for dose) study.” As requested, we have made the following editorial and/or format changes to the manuscript:

- Authors’ Contributions: The authors’ contributions have been placed directly preceding the Acknowledgements. The sample was used to format the authors’ names. We also added the following line to the end of the contributions: “All authors read and approved the final manuscript.”
- Affiliations: We have added “USA” to all of the authors’ affiliations.
- Title Page: We have moved the previous presentation statement to the Acknowledgements section and have added the previous presentations into the References.
- Keywords: The keywords have been deleted from the manuscript text.
- Email Addresses: The authors’ names have been formatted to match the sample provided for email addresses.
- Highlighted/Underlined Text: We have removed all highlighting and underlining from the manuscript.
- Abbreviations: The abbreviations have been formatted into a paragraph.
- Typography: We have reviewed the text for last-minute edits.

Thank you for the opportunity to improve our manuscript. We hope that these revisions meet the Editorial Production Team’s expectations. We look forward to receiving layout proofs for review before publication. Please direct all correspondence concerning this revision to my attention:

Vasundhara Tolia, MD
Adjunct Professor of Pediatrics
Michigan State University
20055 Providence Drive, #400
Southfield, MI 48075
Tel: 248-569-1500
Fax: 248-569-1507
Email: vasu.tolia@gmail.com

Sincerely,

Vasundhara Tolia, MD